Author's response to reviews

Title: Maternal and Neonatal Health Expenditure in Mumbai Slums (India): A Cross Sectional Study

Authors:

Jolene Skordis-Worrall (j.skordis-worrall@ucl.ac.uk)
Noemi Pace (n.pace@unive.it)
Ujwala Bapat (uiwala@snehamumbai.org)
Sushmita Das (sushmita@snehamumbai.org)
Neena S More (neena@snehamumbai.org)
Wasundhara Joshi (wasundhara@snehamumbai.org)
Anni-Maria Pulkki-Brannstrom (a.pulkki-brannstrom@ich.ucl.ac.uk)
David Osrin (d.osrin@ich.ucl.ac.uk)

Version: 4 Date: 21 February 2011

Author’s response to reviews:

Dear Editorial Production Team,

All major and minor revisions have been completed. The changes to the figures (removal of headings) could not be done in the versions sent with the Editorial e-mail, as such these changes have been made on the original .doc files. We hope this is acceptable.

Regards,

Jolene